\xname{H2-Air1}
\W\subsubsection{Cell width vs equivalence ratio; H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air1.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air1.gif}}
\T\caption{Cell width vs equivalence ratio; H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air1.ps}

\T H2-Air1 - Table~\ref{H2-Air1}~\cite[Guirao (1982)]{c_guir82}
\W\xlink{H2-Air1.txt}{\rawdatadir/H2-Air1.txt}
\W Guirao (1982) \cite{c_guir82}
T=293 K, P=101.3 kPa

\T at4 - Table~\ref{at4}~\cite[Tieszen (1986)]{s_ties86}
\W\xlink{at4.txt}{\rawdatadir/at4.txt}
\W Tieszen (1986) \cite{s_ties86}
T=298 K, P=101.3 kPa

\T at14 - Table~\ref{at14}~\cite[Ciccarelli (1994)]{g_cicc94}
\W\xlink{at14.txt}{\rawdatadir/at14.txt}
\W Ciccarelli (1994) \cite{g_cicc94}
T=300 K, P=100 kPa

\T at19 - Table~\ref{at19}~\cite[Benedick (1984)]{w_bene84}
\W\xlink{at19.txt}{\rawdatadir/at19.txt}
\W Benedick (1984) \cite{w_bene84}
T=293 K, P=82.7 kPa

\T\clearpage

\xname{H2-Air2}
\W\subsubsection{Cell width vs equivalence ratio - high temperature; H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air2.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air2.gif}}
\T\caption{Cell width vs equivalence ratio - high temperature; H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air2.ps}

\T at15 - Table~\ref{at15}~\cite[Ciccarelli (1994)]{g_cicc94}
\W\xlink{at15.txt}{\rawdatadir/at15.txt}
\W Ciccarelli (1994) \cite{g_cicc94}
T=500 K, P=100 kPa

\T at16 - Table~\ref{at16}~\cite[Ciccarelli (1994)]{g_cicc94}
\W\xlink{at16.txt}{\rawdatadir/at16.txt}
\W Ciccarelli (1994) \cite{g_cicc94}
T=650 K, P=100 kPa

\T ja20a - Table~\ref{ja20a}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja20a.txt}{\rawdatadir/ja20a.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=300 K, P=100 kPa

\T ja20b - Table~\ref{ja20b}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja20b.txt}{\rawdatadir/ja20b.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=500 K, P=100 kPa

\T ja20c - Table~\ref{ja20c}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja20c.txt}{\rawdatadir/ja20c.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=650 K, P=100 kPa

\T\clearpage

\xname{H2-Air3}
\W\subsubsection{Cell width vs equivalence ratio (constant air density); H2-Air-Steam}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air3.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air3.gif}}
\T\caption{Cell width vs equivalence ratio (constant air density); H2-Air-Steam}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air3.ps}

\T at3a - Table~\ref{at3a}~\cite[Tieszen (1986)]{s_ties86}
\W\xlink{at3a.txt}{\rawdatadir/at3a.txt}
\W Tieszen (1986) \cite{s_ties86}
T=373 K, P=150 - 300 kPa

\T at3b - Table~\ref{at3b}~\cite[Tieszen (1986)]{s_ties86}
\W\xlink{at3b.txt}{\rawdatadir/at3b.txt}
\W Tieszen (1986) \cite{s_ties86}
T=373 K, P=150 - 304 kPa, 10\% steam

\T at3c - Table~\ref{at3c}~\cite[Tieszen (1986)]{s_ties86}
\W\xlink{at3c.txt}{\rawdatadir/at3c.txt}
\W Tieszen (1986) \cite{s_ties86}
T=373 K, P=150 - 304 kPa, 20\% steam

\T at3d - Table~\ref{at3d}~\cite[Tieszen (1986)]{s_ties86}
\W\xlink{at3d.txt}{\rawdatadir/at3d.txt}
\W Tieszen (1986) \cite{s_ties86}
T=373 K, P=150 - 304 kPa, 30\% steam

\T\clearpage

\xname{H2-Air4}
\W\subsubsection{Cell width vs equivalence ratio; H2-Air-CO2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air4.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air4.gif}}
\T\caption{Cell width vs equivalence ratio; H2-Air-CO2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air4.ps}

\T ja26a - Table~\ref{ja26a}~\cite[Tieszen (1987)]{s_ties87}
\W\xlink{ja26a.txt}{\rawdatadir/ja26a.txt}
\W Tieszen (1987) \cite{s_ties87}
T=293 K, P=100 kPa, 5\% CO2

\T ja26b - Table~\ref{ja26b}~\cite[Guirao (1989)]{c_guir89}
\W\xlink{ja26b.txt}{\rawdatadir/ja26b.txt}
\W Guirao (1989) \cite{c_guir89}
T=293 K, P=100 kPa, 5\% CO2

\T ja26c - Table~\ref{ja26c}~\cite[Tieszen (1987)]{s_ties87}
\W\xlink{ja26c.txt}{\rawdatadir/ja26c.txt}
\W Tieszen (1987) \cite{s_ties87}
T=293 K, P=100 kPa, 10\% CO2

\T ja26d - Table~\ref{ja26d}~\cite[Guirao (1989)]{c_guir89}
\W\xlink{ja26d.txt}{\rawdatadir/ja26d.txt}
\W Guirao (1989) \cite{c_guir89}
T=293 K, P=100 kPa, 10\% CO2

\T ja26e - Table~\ref{ja26e}~\cite[Tieszen (1987)]{s_ties87}
\W\xlink{ja26e.txt}{\rawdatadir/ja26e.txt}
\W Tieszen (1987) \cite{s_ties87}
T=293 K, P=100 kPa, 15\% CO2

\T ja26f - Table~\ref{ja26f}~\cite[Guirao (1989)]{c_guir89}
\W\xlink{ja26f.txt}{\rawdatadir/ja26f.txt}
\W Guirao (1989) \cite{c_guir89}
T=293 K, P=100 kPa, 15\% CO2

\T\clearpage

\xname{H2-Air5}
\W\subsubsection{Cell width vs equivalence ratio; H2-Air-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air5.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air5.gif}}
\T\caption{Cell width vs equivalence ratio; H2-Air-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air5.ps}

\T ja25a - Table~\ref{ja25a}~\cite[Stamps (1991)]{d_stam91b}
\W\xlink{ja25a.txt}{\rawdatadir/ja25a.txt}
\W Stamps (1991) \cite{d_stam91b}
T=393 K, P=100 kPa, 30\% H2O

\T ja25b - Table~\ref{ja25b}~\cite[Stamps (1991)]{d_stam91b}
\W\xlink{ja25b.txt}{\rawdatadir/ja25b.txt}
\W Stamps (1991) \cite{d_stam91b}
T=393 K, P=100 kPa, 20\% H2O

\T ja25c - Table~\ref{ja25c}~\cite[Stamps (1991)]{d_stam91b}
\W\xlink{ja25c.txt}{\rawdatadir/ja25c.txt}
\W Stamps (1991) \cite{d_stam91b}
T=393 K, P=100 kPa, 10\% H2O

\T ja25d - Table~\ref{ja25d}~\cite[Stamps (1991)]{d_stam91b}
\W\xlink{ja25d.txt}{\rawdatadir/ja25d.txt}
\W Stamps (1991) \cite{d_stam91b}
T=393 K, P=100 kPa, 20\% CO2

\T ja25e - Table~\ref{ja25e}~\cite[Stamps (1991)]{d_stam91b}
\W\xlink{ja25e.txt}{\rawdatadir/ja25e.txt}
\W Stamps (1991) \cite{d_stam91b}
T=393 K, P=100 kPa, 10\% CO2

\T ja25f - Table~\ref{ja25f}~\cite[Stamps (1991)]{d_stam91b}
\W\xlink{ja25f.txt}{\rawdatadir/ja25f.txt}
\W Stamps (1991) \cite{d_stam91b}
T=393 K, P=100 kPa

\T\clearpage

\xname{H2-Air6}
\W\subsubsection{Cell width vs initial pressure; H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air6.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air6.gif}}
\T\caption{Cell width vs initial pressure; H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air6.ps}

\T at5a - Table~\ref{at5a}~\cite[Stamps (1991)]{d_stam91}
\W\xlink{at5a.txt}{\rawdatadir/at5a.txt}
\W Stamps (1991) \cite{d_stam91}
T=293 K, ER=1

\T at5b - Table~\ref{at5b}~\cite[Stamps (1991)]{d_stam91}
\W\xlink{at5b.txt}{\rawdatadir/at5b.txt}
\W Stamps (1991) \cite{d_stam91}
T=293 K, ER=0.5

\T ja21a - Table~\ref{ja21a}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja21a.txt}{\rawdatadir/ja21a.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=650 K, ER=0.5, 20\% H2O

\T ja21b - Table~\ref{ja21b}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja21b.txt}{\rawdatadir/ja21b.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=650 K, ER=0.4, 30\% H2O

\T ja21c - Table~\ref{ja21c}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja21c.txt}{\rawdatadir/ja21c.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=650 K, ER=0.2

\T\clearpage

\xname{H2-Air7}
\W\subsubsection{Cell width vs initial pressure; H2-Air-Steam}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air7.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air7.gif}}
\T\caption{Cell width vs initial pressure; H2-Air-Steam}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air7.ps}

\T ja23a - Table~\ref{ja23a}~\cite[Stamps (1991)]{d_stam91}
\W\xlink{ja23a.txt}{\rawdatadir/ja23a.txt}
\W Stamps (1991) \cite{d_stam91}
T=373 K, ER=1, 30\% H2O

\T ja23b - Table~\ref{ja23b}~\cite[Tieszen (1986)]{s_ties86}
\W\xlink{ja23b.txt}{\rawdatadir/ja23b.txt}
\W Tieszen (1986) \cite{s_ties86}
T=373 K, ER=1, 30\% H2O

\T ja23c - Table~\ref{ja23c}~\cite[Stamps (1991)]{d_stam91}
\W\xlink{ja23c.txt}{\rawdatadir/ja23c.txt}
\W Stamps (1991) \cite{d_stam91}
T=373 K, ER=1, 20\% H2O

\T ja23d - Table~\ref{ja23d}~\cite[Tieszen (1986)]{s_ties86}
\W\xlink{ja23d.txt}{\rawdatadir/ja23d.txt}
\W Tieszen (1986) \cite{s_ties86}
T=373 K, ER=1, 20\% H2O

\T ja23e - Table~\ref{ja23e}~\cite[Stamps (1991)]{d_stam91}
\W\xlink{ja23e.txt}{\rawdatadir/ja23e.txt}
\W Stamps (1991) \cite{d_stam91}
T=373 K, ER=1, 10\% H2O

\T ja23f - Table~\ref{ja23f}~\cite[Tieszen (1986)]{s_ties86}
\W\xlink{ja23f.txt}{\rawdatadir/ja23f.txt}
\W Tieszen (1986) \cite{s_ties86}
T=373 K, ER=1, 10\% H2O

\T ja23g - Table~\ref{ja23g}~\cite[Stamps (1991)]{d_stam91}
\W\xlink{ja23g.txt}{\rawdatadir/ja23g.txt}
\W Stamps (1991) \cite{d_stam91}
T=373 K, ER=1

\T ja23h - Table~\ref{ja23h}~\cite[Tieszen (1986)]{s_ties86}
\W\xlink{ja23h.txt}{\rawdatadir/ja23h.txt}
\W Tieszen (1986) \cite{s_ties86}
T=373 K, ER=1

\T\clearpage

\xname{H2-Air8}
\W\subsubsection{Cell width vs. percent diluent; H2-Air-Steam}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air8.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air8.gif}}
\T\caption{Cell width vs. percent diluent; H2-Air-Steam}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air8.ps}

\T ja22a - Table~\ref{ja22a}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja22a.txt}{\rawdatadir/ja22a.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=650 K, P=100 kPa, ER=1

\T ja22b - Table~\ref{ja22b}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja22b.txt}{\rawdatadir/ja22b.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=400 K, P=100 kPa, ER=1

\T ja22c - Table~\ref{ja22c}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja22c.txt}{\rawdatadir/ja22c.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=650 K, P=100 kPa, ER=0.5

\T ja22d - Table~\ref{ja22d}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja22d.txt}{\rawdatadir/ja22d.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=400 K, P=100 kPa, ER=0.5

\T ja22e - Table~\ref{ja22e}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja22e.txt}{\rawdatadir/ja22e.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
T=650 K, P=100 kPa, ER=0.3

\T\clearpage

\xname{H2-Air9}
\W\subsubsection{Cell width vs. initial temperature; H2-Air, H2-Air-Steam}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air9.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air9.gif}}
\T\caption{Cell width vs. initial temperature; H2-Air, H2-Air-Steam}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air9.ps}

\T ja22f - Table~\ref{ja22f}~\cite[Ciccarelli (1997)]{g_cicc97}
\W\xlink{ja22f.txt}{\rawdatadir/ja22f.txt}
\W Ciccarelli (1997) \cite{g_cicc97}
P=100 kPa, ER=1, 30\% H2O

\T ja27a - Table~\ref{ja27a}~\cite[Tieszen (1987)]{s_ties87}
\W\xlink{ja27a.txt}{\rawdatadir/ja27a.txt}
\W Tieszen (1987) \cite{s_ties87}
P=100 kPa, ER=0.448-0.51

\T ja27b - Table~\ref{ja27b}~\cite[Tieszen (1987)]{s_ties87}
\W\xlink{ja27b.txt}{\rawdatadir/ja27b.txt}
\W Tieszen (1987) \cite{s_ties87}
P=100-102 kPa, ER=0.446-0.511

\T\clearpage

\xname{H2-N2Ox1}
\W\subsubsection{Cell width vs equivalence ratio; H2-N2O-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-N2Ox1.ps,width=6in}}
{\htmlimage{\plotdir/H2-N2Ox1.gif}}
\T\caption{Cell width vs equivalence ratio; H2-N2O-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-N2Ox1.ps}

\T ja5a - Table~\ref{ja5a}~\cite[Pfahl (1998)]{u_pfah98}
\W\xlink{ja5a.txt}{\rawdatadir/ja5a.txt}
\W Pfahl (1998) \cite{u_pfah98}
T=293 K, P=70.9 kPa

\T ja5b - Table~\ref{ja5b}~\cite[Pfahl (1998)]{u_pfah98}
\W\xlink{ja5b.txt}{\rawdatadir/ja5b.txt}
\W Pfahl (1998) \cite{u_pfah98}
T=293 K, P=70.9 kPa, 30\% N2

\T ja5c - Table~\ref{ja5c}~\cite[Pfahl (1998)]{u_pfah98}
\W\xlink{ja5c.txt}{\rawdatadir/ja5c.txt}
\W Pfahl (1998) \cite{u_pfah98}
T=293 K, P=70.9 kPa, 54.5\% N2

\T\clearpage

\xname{H2-N2Ox2}
\W\subsubsection{Cell width vs percent diluent; H2-N2O-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-N2Ox2.ps,width=6in}}
{\htmlimage{\plotdir/H2-N2Ox2.gif}}
\T\caption{Cell width vs percent diluent; H2-N2O-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-N2Ox2.ps}

\T ja5d - Table~\ref{ja5d}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja5d.txt}{\rawdatadir/ja5d.txt}
\W Akbar (1997) \cite{r_akbar97}
P=100 kPa, ER=1, 50-70\% N2, Oxidizer=N2O

\T ja5e - Table~\ref{ja5e}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja5e.txt}{\rawdatadir/ja5e.txt}
\W Akbar (1997) \cite{r_akbar97}
P=70-100 kPa, ER=0.07-0.39, 28-76\% Air, Oxidizer=N2O

\T\clearpage

\xname{H2-Ox1}
\W\subsubsection{Cell width vs initial pressure; H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox1.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox1.gif}}
\T\caption{Cell width vs initial pressure; H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox1.ps}

\T at62a - Table~\ref{at62a}~\cite[Strehlow (1969)]{r_stre69b}
\W\xlink{at62a.txt}{\rawdatadir/at62a.txt}
\W Strehlow (1969) \cite{r_stre69b}
T=293 K, ER=1

\T at57a - Table~\ref{at57a}~\cite[Manzhalei (1974)]{v_manz74}
\W\xlink{at57a.txt}{\rawdatadir/at57a.txt}
\W Manzhalei (1974) \cite{v_manz74}
T=293 K, ER=1

\T mk9b - Table~\ref{mk9b}~\cite[Denisov (1960)]{y_deni60}
\W\xlink{mk9b.txt}{\rawdatadir/mk9b.txt}
\W Denisov (1960) \cite{y_deni60}
T=293 K, ER=1

\T at21a - Table~\ref{at21a}~\cite[Desbordes (1990)]{d_desb90}
\W\xlink{at21a.txt}{\rawdatadir/at21a.txt}
\W Desbordes (1990) \cite{d_desb90}
T=293 K, ER=1

\T at21b - Table~\ref{at21b}~\cite[Zitoun (1995)]{r_zito95}
\W\xlink{at21b.txt}{\rawdatadir/at21b.txt}
\W Zitoun (1995) \cite{r_zito95}
T=293 K, ER=1

\T\clearpage

\xname{H2-Ox2}
\W\subsubsection{Cell width vs initial pressure; H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox2.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox2.gif}}
\T\caption{Cell width vs initial pressure; H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox2.ps}

\T at21c - Table~\ref{at21c}~\cite[Zitoun (1995)]{r_zito95}
\W\xlink{at21c.txt}{\rawdatadir/at21c.txt}
\W Zitoun (1995) \cite{r_zito95}
T=123 K, ER=1

\T at47a - Table~\ref{at47a}~\cite[Lee (1977)]{j_lee77}
\W\xlink{at47a.txt}{\rawdatadir/at47a.txt}
\W Lee (1977) \cite{j_lee77}
T=293 K, ER=1

\T at58a - Table~\ref{at58a}~\cite[Barthel (1974)]{h_bart74}
\W\xlink{at58a.txt}{\rawdatadir/at58a.txt}
\W Barthel (1974) \cite{h_bart74}
T=298 K, ER=1

\T at74 - Table~\ref{at74}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at74.txt}{\rawdatadir/at74.txt}
\W Knystautas (1982) \cite{r_knys82}
T=293 K, ER=1

\T mk8a - Table~\ref{mk8a}~\cite[Voitsekhovskii (1966)]{b_voit66}
\W\xlink{mk8a.txt}{\rawdatadir/mk8a.txt}
\W Voitsekhovskii (1966) \cite{b_voit66}
T=293 K, ER=1

\T mk9b - Table~\ref{mk9b}~\cite[Denisov (1960)]{y_deni60}
\W\xlink{mk9b.txt}{\rawdatadir/mk9b.txt}
\W Denisov (1960) \cite{y_deni60}
T=293 K, ER=1

\T\clearpage

\xname{H2-Ox3}
\W\subsubsection{Cell width vs initial pressure (stoichiometric); H2-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox3.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox3.gif}}
\T\caption{Cell width vs initial pressure (stoichiometric); H2-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox3.ps}

\T ja1c - Table~\ref{ja1c}~\cite[Kaneshige (1999)]{m_kane99}
\W\xlink{ja1c.txt}{\rawdatadir/ja1c.txt}
\W Kaneshige (1999) \cite{m_kane99}
T=293 K, ER=1, 25\% N2

\T ja1d - Table~\ref{ja1d}~\cite[Kaneshige (1999)]{m_kane99}
\W\xlink{ja1d.txt}{\rawdatadir/ja1d.txt}
\W Kaneshige (1999) \cite{m_kane99}
T=293 K, ER=1, 40\% N2

\T ja1e - Table~\ref{ja1e}~\cite[Kaneshige (1999)]{m_kane99}
\W\xlink{ja1e.txt}{\rawdatadir/ja1e.txt}
\W Kaneshige (1999) \cite{m_kane99}
T=293 K, ER=1, 50\% N2

\T\clearpage

\xname{H2-Ox4}
\W\subsubsection{Cell width vs initial pressure (stoichiometric) - part1; H2-O2-Ar}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox4.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox4.gif}}
\T\caption{Cell width vs initial pressure (stoichiometric) - part1; H2-O2-Ar}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox4.ps}

\T at62b - Table~\ref{at62b}~\cite[Strehlow (1969)]{r_stre69b}
\W\xlink{at62b.txt}{\rawdatadir/at62b.txt}
\W Strehlow (1969) \cite{r_stre69b}
T=293 K, ER=1, 40\% Ar

\T at69d - Table~\ref{at69d}~\cite[Strehlow (1967)]{r_stre67}
\W\xlink{at69d.txt}{\rawdatadir/at69d.txt}
\W Strehlow (1967) \cite{r_stre67}
T=293 K, ER=1, 70\% Ar

\T at58b - Table~\ref{at58b}~\cite[Barthel (1974)]{h_bart74}
\W\xlink{at58b.txt}{\rawdatadir/at58b.txt}
\W Barthel (1974) \cite{h_bart74}
T=298 K, ER=1, 40\% Ar

\T at33a - Table~\ref{at33a}~\cite[Anderson (1992)]{t_ande92}
\W\xlink{at33a.txt}{\rawdatadir/at33a.txt}
\W Anderson (1992) \cite{t_ande92}
T=293 K, ER=1, 55.6\% Ar

\T at33b - Table~\ref{at33b}~\cite[Anderson (1992)]{t_ande92}
\W\xlink{at33b.txt}{\rawdatadir/at33b.txt}
\W Anderson (1992) \cite{t_ande92}
T=293 K, ER=1, 50\% Ar

\T\clearpage

\xname{H2-Ox5}
\W\subsubsection{Cell width vs initial pressure (stoichiometric) - part2; H2-O2-Ar}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox5.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox5.gif}}
\T\caption{Cell width vs initial pressure (stoichiometric) - part2; H2-O2-Ar}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox5.ps}

\T at69d - Table~\ref{at69d}~\cite[Strehlow (1967)]{r_stre67}
\W\xlink{at69d.txt}{\rawdatadir/at69d.txt}
\W Strehlow (1967) \cite{r_stre67}
T=293 K, ER=1, 70\% Ar

\T at69e - Table~\ref{at69e}~\cite[Strehlow (1967)]{r_stre67}
\W\xlink{at69e.txt}{\rawdatadir/at69e.txt}
\W Strehlow (1967) \cite{r_stre67}
T=293 K, ER=1, 77.5\% Ar

\T at69g - Table~\ref{at69g}~\cite[Strehlow (1967)]{r_stre67}
\W\xlink{at69g.txt}{\rawdatadir/at69g.txt}
\W Strehlow (1967) \cite{r_stre67}
T=293 K, ER=1, 85\% Ar

\T at58c - Table~\ref{at58c}~\cite[Barthel (1974)]{h_bart74}
\W\xlink{at58c.txt}{\rawdatadir/at58c.txt}
\W Barthel (1974) \cite{h_bart74}
T=298 K, ER=1, 70\% Ar

\T at47b - Table~\ref{at47b}~\cite[Lee (1977)]{j_lee77}
\W\xlink{at47b.txt}{\rawdatadir/at47b.txt}
\W Lee (1977) \cite{j_lee77}
T=293 K, ER=1, 85\% Ar

\T at65 - Table~\ref{at65}~\cite[Strehlow (1969)]{r_stre69c}
\W\xlink{at65.txt}{\rawdatadir/at65.txt}
\W Strehlow (1969) \cite{r_stre69c}
T=600 - 800 K, ER=1, 70\% Ar

\T\clearpage

\xname{H2-Ox6}
\W\subsubsection{Cell width vs initial pressure; H2-O2-Ar}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox6.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox6.gif}}
\T\caption{Cell width vs initial pressure; H2-O2-Ar}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox6.ps}

\T at64c - Table~\ref{at64c}~\cite[Strehlow (1969)]{r_stre69c}
\W\xlink{at64c.txt}{\rawdatadir/at64c.txt}
\W Strehlow (1969) \cite{r_stre69c}
T=293 K, ER=0.36, 60\% Ar

\T at64b - Table~\ref{at64b}~\cite[Strehlow (1969)]{r_stre69c}
\W\xlink{at64b.txt}{\rawdatadir/at64b.txt}
\W Strehlow (1969) \cite{r_stre69c}
T=293 K, ER=2.12, 60\% Ar

\T at33c - Table~\ref{at33c}~\cite[Anderson (1992)]{t_ande92}
\W\xlink{at33c.txt}{\rawdatadir/at33c.txt}
\W Anderson (1992) \cite{t_ande92}
T=293 K, ER=0.75, 50\% Ar

\T\clearpage

\xname{H2-Ox7}
\W\subsubsection{Cell width vs initial pressure; H2-O2-He, H2-O2-N2-Ar}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox7.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox7.gif}}
\T\caption{Cell width vs initial pressure; H2-O2-He, H2-O2-N2-Ar}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox7.ps}

\T at69f - Table~\ref{at69f}~\cite[Strehlow (1967)]{r_stre67}
\W\xlink{at69f.txt}{\rawdatadir/at69f.txt}
\W Strehlow (1967) \cite{r_stre67}
T=293 K, ER=1, 70\% He

\T at64a - Table~\ref{at64a}~\cite[Strehlow (1969)]{r_stre69c}
\W\xlink{at64a.txt}{\rawdatadir/at64a.txt}
\W Strehlow (1969) \cite{r_stre69c}
T=293 K, ER=1, 76\% 3.75Ar+N2

\T at42a - Table~\ref{at42a}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at42a.txt}{\rawdatadir/at42a.txt}
\W Kumar (1990) \cite{r_kuma90}
T=295 K, ER=1, 55\% He

\T at42b - Table~\ref{at42b}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at42b.txt}{\rawdatadir/at42b.txt}
\W Kumar (1990) \cite{r_kuma90}
T=295 K, ER=1, 70\% He

\T\clearpage

\xname{H2-Ox8}
\W\subsubsection{Cell width vs initial pressure; H2-Cl2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox8.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox8.gif}}
\T\caption{Cell width vs initial pressure; H2-Cl2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox8.ps}

\T at182a - Table~\ref{at182a}~\cite[Knystautas (1988)]{r_knys88}
\W\xlink{at182a.txt}{\rawdatadir/at182a.txt}
\W Knystautas (1988) \cite{r_knys88}
ER=0.56

\T at182b - Table~\ref{at182b}~\cite[Knystautas (1988)]{r_knys88}
\W\xlink{at182b.txt}{\rawdatadir/at182b.txt}
\W Knystautas (1988) \cite{r_knys88}
ER=0.67

\T at182c - Table~\ref{at182c}~\cite[Knystautas (1988)]{r_knys88}
\W\xlink{at182c.txt}{\rawdatadir/at182c.txt}
\W Knystautas (1988) \cite{r_knys88}
ER=1.5

\T at182d - Table~\ref{at182d}~\cite[Knystautas (1988)]{r_knys88}
\W\xlink{at182d.txt}{\rawdatadir/at182d.txt}
\W Knystautas (1988) \cite{r_knys88}
ER=1

\T\clearpage

\xname{H2-Ox9}
\W\subsubsection{Cell width vs percent diluent; H2-O2-He-CO2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox9.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox9.gif}}
\T\caption{Cell width vs percent diluent; H2-O2-He-CO2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox9.ps}

\T at38a - Table~\ref{at38a}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at38a.txt}{\rawdatadir/at38a.txt}
\W Kumar (1990) \cite{r_kuma90}
T=295 K, P=106.6 kPa, ER=1, 30-55\% He and CO2, Fuel=H2

\T at38b - Table~\ref{at38b}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at38b.txt}{\rawdatadir/at38b.txt}
\W Kumar (1990) \cite{r_kuma90}
T=295 K, P=106.6 kPa, ER=1, 40-65\% He and CO2, Fuel=H2

\T at38c - Table~\ref{at38c}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at38c.txt}{\rawdatadir/at38c.txt}
\W Kumar (1990) \cite{r_kuma90}
T=295 K, P=106.6 kPa, ER=1, 35-70\% He and CO2, Fuel=H2

\T\clearpage

\xname{H2-Ox10}
\W\subsubsection{Cell width vs percent diluent; H2-O2-He-H2O}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox10.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox10.gif}}
\T\caption{Cell width vs percent diluent; H2-O2-He-H2O}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox10.ps}

\T at39a - Table~\ref{at39a}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at39a.txt}{\rawdatadir/at39a.txt}
\W Kumar (1990) \cite{r_kuma90}
T=353 K, P=106.6 kPa, ER=1, 5-70\% He and H2O, Fuel=H2

\T at39b - Table~\ref{at39b}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at39b.txt}{\rawdatadir/at39b.txt}
\W Kumar (1990) \cite{r_kuma90}
T=353 K, P=106.6 kPa, ER=1, 5-70\% He and H2O, Fuel=H2

\T at39c - Table~\ref{at39c}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at39c.txt}{\rawdatadir/at39c.txt}
\W Kumar (1990) \cite{r_kuma90}
T=353 K, P=106.6 kPa, ER=1, 5-70\% He and H2O, Fuel=H2

\T at39d - Table~\ref{at39d}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at39d.txt}{\rawdatadir/at39d.txt}
\W Kumar (1990) \cite{r_kuma90}
T=353 K, P=106.6 kPa, ER=1, 5-70\% He and H2O, Fuel=H2

\T\clearpage

\xname{H2-Ox11}
\W\subsubsection{Cell width vs percent diluent; H2-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox11.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox11.gif}}
\T\caption{Cell width vs percent diluent; H2-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox11.ps}

\T mk7b - Table~\ref{mk7b}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{mk7b.txt}{\rawdatadir/mk7b.txt}
\W Knystautas (1982) \cite{r_knys82}
T=293 K, P=101.3 kPa, ER=1, Fuel=H2

\T ja1a - Table~\ref{ja1a}~\cite[Kaneshige (1999)]{m_kane99}
\W\xlink{ja1a.txt}{\rawdatadir/ja1a.txt}
\W Kaneshige (1999) \cite{m_kane99}
T=293 K, P=100 kPa, ER=1, Fuel=H2

\T ja1b - Table~\ref{ja1b}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja1b.txt}{\rawdatadir/ja1b.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, P=100 kPa, ER=1, Fuel=H2

\T\clearpage

\xname{H2-Ox12}
\W\subsubsection{Cell width vs percent diluent; H2-O2-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox12.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox12.gif}}
\T\caption{Cell width vs percent diluent; H2-O2-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox12.ps}

\T at35a - Table~\ref{at35a}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at35a.txt}{\rawdatadir/at35a.txt}
\W Kumar (1990) \cite{r_kuma90}
T=293 K, P=106.6 kPa, ER=1, 25-85\% He, Fuel=H2

\T at35b - Table~\ref{at35b}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at35b.txt}{\rawdatadir/at35b.txt}
\W Kumar (1990) \cite{r_kuma90}
T=293 K, P=106.6 kPa, ER=1, 25-85\% Ar, Fuel=H2

\T at37 - Table~\ref{at37}~\cite[Kumar (1990)]{r_kuma90}
\W\xlink{at37.txt}{\rawdatadir/at37.txt}
\W Kumar (1990) \cite{r_kuma90}
T=293 K, P=106.6 kPa, ER=1, 25-85\% He, Fuel=Deuterium

\T ja2 - Table~\ref{ja2}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja2.txt}{\rawdatadir/ja2.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, P=100 kPa, ER=1, 57-87\% Ar, Fuel=H2

\T ja3 - Table~\ref{ja3}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja3.txt}{\rawdatadir/ja3.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, P=100 kPa, ER=1, 20-90\% He, Fuel=H2

\T ja4 - Table~\ref{ja4}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja4.txt}{\rawdatadir/ja4.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, P=100 kPa, ER=1, 10-40\% CO2, Fuel=H2

\T\clearpage

\xname{H2-Ox13}
\W\subsubsection{Cell length vs initial pressure; CO-H2-O2-Ar-Inhibitor}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox13.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox13.gif}}
\T\caption{Cell length vs initial pressure; CO-H2-O2-Ar-Inhibitor}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox13.ps}

\T at169a - Table~\ref{at169a}~\cite[Libouton (1975)]{j_libo75}
\W\xlink{at169a.txt}{\rawdatadir/at169a.txt}
\W Libouton (1975) \cite{j_libo75}
ER=1, 30\% Ar

\T at169b - Table~\ref{at169b}~\cite[Libouton (1975)]{j_libo75}
\W\xlink{at169b.txt}{\rawdatadir/at169b.txt}
\W Libouton (1975) \cite{j_libo75}
ER=1, 30\% 11.88Ar+CF3Cl

\T at169c - Table~\ref{at169c}~\cite[Libouton (1975)]{j_libo75}
\W\xlink{at169c.txt}{\rawdatadir/at169c.txt}
\W Libouton (1975) \cite{j_libo75}
ER=1, 30\% 11.88Ar+CF2Cl2

\T at169d - Table~\ref{at169d}~\cite[Libouton (1975)]{j_libo75}
\W\xlink{at169d.txt}{\rawdatadir/at169d.txt}
\W Libouton (1975) \cite{j_libo75}
ER=1, 30\% 11.88Ar+CFCl3

\T at169e - Table~\ref{at169e}~\cite[Libouton (1975)]{j_libo75}
\W\xlink{at169e.txt}{\rawdatadir/at169e.txt}
\W Libouton (1975) \cite{j_libo75}
ER=1, 30\% 11.88Ar+CF3Br

\T\clearpage

\xname{H2-Ox14}
\W\subsubsection{Cell length vs percent inhibitor; H2-O2-Ar-Inhibitor}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox14.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox14.gif}}
\T\caption{Cell length vs percent inhibitor; H2-O2-Ar-Inhibitor}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox14.ps}

\T at70a - Table~\ref{at70a}~\cite[Nzeyimana (1991)]{e_nzey91}
\W\xlink{at70a.txt}{\rawdatadir/at70a.txt}
\W Nzeyimana (1991) \cite{e_nzey91}
P=26.7 kPa, ER=0.8, 50\% Ar+CF4

\T at70b - Table~\ref{at70b}~\cite[Nzeyimana (1991)]{e_nzey91}
\W\xlink{at70b.txt}{\rawdatadir/at70b.txt}
\W Nzeyimana (1991) \cite{e_nzey91}
P=26.7 kPa, ER=1, 50\% Ar+CF4

\T at70c - Table~\ref{at70c}~\cite[Nzeyimana (1991)]{e_nzey91}
\W\xlink{at70c.txt}{\rawdatadir/at70c.txt}
\W Nzeyimana (1991) \cite{e_nzey91}
P=26.7 kPa, ER=1.2, 50\% Ar+CF4

\T at189 - Table~\ref{at189}~\cite[Lefebvre (1993)]{m_lefe93}
\W\xlink{at189.txt}{\rawdatadir/at189.txt}
\W Lefebvre (1993) \cite{m_lefe93}
P=26.7 kPa, ER=1, 50\% Ar+CF3H

\T\clearpage

\xname{CH4-Air1}
\W\subsubsection{Cell width vs initial pressure; CH4-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Air1.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Air1.gif}}
\T\caption{Cell width vs initial pressure; CH4-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Air1.ps}

\T at157a - Table~\ref{at157a}~\cite[Moen (1984)]{i_moen84}
\W\xlink{at157a.txt}{\rawdatadir/at157a.txt}
\W Moen (1984) \cite{i_moen84}
T=293 K, ER=1

\T at13a - Table~\ref{at13a}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{at13a.txt}{\rawdatadir/at13a.txt}
\W Knystautas (1984) \cite{r_knys84}
T=293 K, ER=1

\T at176a - Table~\ref{at176a}~\cite[Beeson (1991)]{h_bees91}
\W\xlink{at176a.txt}{\rawdatadir/at176a.txt}
\W Beeson (1991) \cite{h_bees91}
T=293 K, ER=1

\T\clearpage

\xname{CH4-Ox1}
\W\subsubsection{Cell width vs initial pressure; CH4-O2-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox1.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox1.gif}}
\T\caption{Cell width vs initial pressure; CH4-O2-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox1.ps}

\T at199a - Table~\ref{at199a}~\cite[Laberge (1993)]{s_labe93}
\W\xlink{at199a.txt}{\rawdatadir/at199a.txt}
\W Laberge (1993) \cite{s_labe93}
T=293 K, ER=1

\T at128a - Table~\ref{at128a}~\cite[Strehlow (1969)]{r_stre69b}
\W\xlink{at128a.txt}{\rawdatadir/at128a.txt}
\W Strehlow (1969) \cite{r_stre69b}
T=293 K, ER=1, 50\% Ar

\T at192a - Table~\ref{at192a}~\cite[Abid (1991)]{s_abid91}
\W\xlink{at192a.txt}{\rawdatadir/at192a.txt}
\W Abid (1991) \cite{s_abid91}
T=293 K, ER=1

\T at184a - Table~\ref{at184a}~\cite[Pedley (1988)]{m_pedl88}
\W\xlink{at184a.txt}{\rawdatadir/at184a.txt}
\W Pedley (1988) \cite{m_pedl88}
T=293 K, ER=1

\T at57d - Table~\ref{at57d}~\cite[Manzhalei (1974)]{v_manz74}
\W\xlink{at57d.txt}{\rawdatadir/at57d.txt}
\W Manzhalei (1974) \cite{v_manz74}
T=293 K, ER=1

\T at73 - Table~\ref{at73}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at73.txt}{\rawdatadir/at73.txt}
\W Knystautas (1982) \cite{r_knys82}
T=293 K, ER=1

\T\clearpage

\xname{CH4-Ox2}
\W\subsubsection{Cell width vs equivalence ratio; CH4-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox2.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox2.gif}}
\T\caption{Cell width vs equivalence ratio; CH4-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox2.ps}

\T at193b - Table~\ref{at193b}~\cite[Aminallah (1993)]{m_amin93}
\W\xlink{at193b.txt}{\rawdatadir/at193b.txt}
\W Aminallah (1993) \cite{m_amin93}
T=293 K, P=120 kPa

\T\clearpage

\xname{CH4-Ox3}
\W\subsubsection{Cell width vs. percent diluent; CH4-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox3.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox3.gif}}
\T\caption{Cell width vs. percent diluent; CH4-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox3.ps}

\T ja6a - Table~\ref{ja6a}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja6a.txt}{\rawdatadir/ja6a.txt}
\W Akbar (1997) \cite{r_akbar97}
T=293 K, P=70-72 kPa

\T ja6b - Table~\ref{ja6b}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja6b.txt}{\rawdatadir/ja6b.txt}
\W Akbar (1997) \cite{r_akbar97}
T=293 K, P=102 kPa

\T\clearpage

\xname{CH4-Ox4}
\W\subsubsection{Cell length vs equivalence ratio; CH4-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox4.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox4.gif}}
\T\caption{Cell length vs equivalence ratio; CH4-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox4.ps}

\T at193a - Table~\ref{at193a}~\cite[Aminallah (1993)]{m_amin93}
\W\xlink{at193a.txt}{\rawdatadir/at193a.txt}
\W Aminallah (1993) \cite{m_amin93}
T=293 K, P=120 kPa

\T\clearpage

\xname{CH4-Ox5}
\W\subsubsection{Cell length vs initial pressure; CH4-O2-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox5.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox5.gif}}
\T\caption{Cell length vs initial pressure; CH4-O2-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox5.ps}

\T at194a - Table~\ref{at194a}~\cite[Bauer (1985)]{p_baue85}
\W\xlink{at194a.txt}{\rawdatadir/at194a.txt}
\W Bauer (1985) \cite{p_baue85}
T=293 K, ER=1

\T at194b - Table~\ref{at194b}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at194b.txt}{\rawdatadir/at194b.txt}
\W Bull (1982) \cite{d_bull82b}
T=293 K, ER=1

\T at194c - Table~\ref{at194c}~\cite[Aminallah (1993)]{m_amin93}
\W\xlink{at194c.txt}{\rawdatadir/at194c.txt}
\W Aminallah (1993) \cite{m_amin93}
T=293 K, ER=1

\T mk5a - Table~\ref{mk5a}~\cite[Bauer (1986)]{p_baue86}
\W\xlink{mk5a.txt}{\rawdatadir/mk5a.txt}
\W Bauer (1986) \cite{p_baue86}
T=293 K, ER=1.08, 31.9\% N2

\T mk5b - Table~\ref{mk5b}~\cite[Bauer (1986)]{p_baue86}
\W\xlink{mk5b.txt}{\rawdatadir/mk5b.txt}
\W Bauer (1986) \cite{p_baue86}
T=293 K, ER=1.09, 55.0\% N2

\T mk5c - Table~\ref{mk5c}~\cite[Bauer (1986)]{p_baue86}
\W\xlink{mk5c.txt}{\rawdatadir/mk5c.txt}
\W Bauer (1986) \cite{p_baue86}
T=293 K, ER=1.15, 64.2\% N2

\T\clearpage

\xname{CH4-N2O}
\W\subsubsection{Cell width vs. percent diluent; CH4-N2O-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-N2O.ps,width=6in}}
{\htmlimage{\plotdir/CH4-N2O.gif}}
\T\caption{Cell width vs. percent diluent; CH4-N2O-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-N2O.ps}

\T ja7a - Table~\ref{ja7a}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja7a.txt}{\rawdatadir/ja7a.txt}
\W Akbar (1997) \cite{r_akbar97}
T=293 K, P=57-72 kPa, ER=1, 0-50\% N2

\T ja7b - Table~\ref{ja7b}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja7b.txt}{\rawdatadir/ja7b.txt}
\W Akbar (1997) \cite{r_akbar97}
T=293 K, P=77-87 kPa, ER=1, 37.5-50\% N2

\T ja7c - Table~\ref{ja7c}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja7c.txt}{\rawdatadir/ja7c.txt}
\W Akbar (1997) \cite{r_akbar97}
T=293 K, P=92-97 kPa, ER=1, 54.5-58.3\% N2

\T ja7d - Table~\ref{ja7d}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja7d.txt}{\rawdatadir/ja7d.txt}
\W Akbar (1997) \cite{r_akbar97}
T=293 K, P=102 kPa, ER=1, 61.5-64.3\% N2

\T ja7e - Table~\ref{ja7e}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja7e.txt}{\rawdatadir/ja7e.txt}
\W Akbar (1997) \cite{r_akbar97}
T=293 K, P=86-97 kPa, ER=0.5-0.7, 48.8-65.1\% Air

\T\clearpage

\xname{C2H2-Ox1a}
\W\subsubsection{Cell width vs initial pressure (stoichiometric) - part 1; C2H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox1a.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox1a.gif}}
\T\caption{Cell width vs initial pressure (stoichiometric) - part 1; C2H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox1a.ps}

\T at198d - Table~\ref{at198d}~\cite[Desbordes (1993)]{d_desb93}
\W\xlink{at198d.txt}{\rawdatadir/at198d.txt}
\W Desbordes (1993) \cite{d_desb93}
T=293 K, ER=1

\T at187a - Table~\ref{at187a}~\cite[Desbordes (1988)]{d_desb88}
\W\xlink{at187a.txt}{\rawdatadir/at187a.txt}
\W Desbordes (1988) \cite{d_desb88}
T=293 K, ER=1

\T at166a - Table~\ref{at166a}~\cite[Edwards (1978)]{d_edwa78}
\W\xlink{at166a.txt}{\rawdatadir/at166a.txt}
\W Edwards (1978) \cite{d_edwa78}
T=293 K, ER=1

\T at166b - Table~\ref{at166b}~\cite[Desbordes (1986)]{d_desb86b}
\W\xlink{at166b.txt}{\rawdatadir/at166b.txt}
\W Desbordes (1986) \cite{d_desb86b}
T=293 K, ER=1

\T at57f - Table~\ref{at57f}~\cite[Manzhalei (1974)]{v_manz74}
\W\xlink{at57f.txt}{\rawdatadir/at57f.txt}
\W Manzhalei (1974) \cite{v_manz74}
T=293 K, ER=1

\T\clearpage

\xname{C2H2-Ox1b}
\W\subsubsection{Cell width vs initial pressure (stoichiometric) - part 2; C2H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox1b.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox1b.gif}}
\T\caption{Cell width vs initial pressure (stoichiometric) - part 2; C2H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox1b.ps}

\T at199c - Table~\ref{at199c}~\cite[Laberge (1993)]{s_labe93}
\W\xlink{at199c.txt}{\rawdatadir/at199c.txt}
\W Laberge (1993) \cite{s_labe93}
T=293 K, ER=1

\T at129c - Table~\ref{at129c}~\cite[Strehlow (1969)]{r_stre69b}
\W\xlink{at129c.txt}{\rawdatadir/at129c.txt}
\W Strehlow (1969) \cite{r_stre69b}
T=293 K, ER=1

\T at148a - Table~\ref{at148a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at148a.txt}{\rawdatadir/at148a.txt}
\W Knystautas (1982) \cite{r_knys82}
T=293 K, ER=1

\T mk8b - Table~\ref{mk8b}~\cite[Voitsekhovskii (1966)]{b_voit66}
\W\xlink{mk8b.txt}{\rawdatadir/mk8b.txt}
\W Voitsekhovskii (1966) \cite{b_voit66}
T=293 K, ER=1

\T mk9a - Table~\ref{mk9a}~\cite[Denisov (1960)]{y_deni60}
\W\xlink{mk9a.txt}{\rawdatadir/mk9a.txt}
\W Denisov (1960) \cite{y_deni60}
T=293 K, ER=1

\T\clearpage

\xname{C2H2-Ox2}
\W\subsubsection{Cell width vs initial pressure; C2H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox2.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox2.gif}}
\T\caption{Cell width vs initial pressure; C2H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox2.ps}

\T at199d - Table~\ref{at199d}~\cite[Laberge (1993)]{s_labe93}
\W\xlink{at199d.txt}{\rawdatadir/at199d.txt}
\W Laberge (1993) \cite{s_labe93}
T=293 K, ER=2.5

\T at129e - Table~\ref{at129e}~\cite[Strehlow (1969)]{r_stre69b}
\W\xlink{at129e.txt}{\rawdatadir/at129e.txt}
\W Strehlow (1969) \cite{r_stre69b}
T=293 K, ER=0.625

\T at150a - Table~\ref{at150a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at150a.txt}{\rawdatadir/at150a.txt}
\W Knystautas (1982) \cite{r_knys82}
T=293 K, ER=2.5

\T at150b - Table~\ref{at150b}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at150b.txt}{\rawdatadir/at150b.txt}
\W Knystautas (1982) \cite{r_knys82}
T=293 K, ER=2.5

\T at151 - Table~\ref{at151}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at151.txt}{\rawdatadir/at151.txt}
\W Knystautas (1982) \cite{r_knys82}
T=293 K, ER=0.625

\T at57c-mk - Table~\ref{at57c-mk}~\cite[Manzhalei (1974)]{v_manz74}
\W\xlink{at57c-mk.txt}{\rawdatadir/at57c-mk.txt}
\W Manzhalei (1974) \cite{v_manz74}
T=293 K, ER=2.5

\T\clearpage

\xname{C2H2-Ox3}
\W\subsubsection{Cell width vs initial pressure; C2H2-O2-Ar}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox3.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox3.gif}}
\T\caption{Cell width vs initial pressure; C2H2-O2-Ar}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox3.ps}

\T at197b - Table~\ref{at197b}~\cite[Desbordes (1993)]{d_desb93}
\W\xlink{at197b.txt}{\rawdatadir/at197b.txt}
\W Desbordes (1993) \cite{d_desb93}
ER=1, 75\% Ar

\T at198b - Table~\ref{at198b}~\cite[Desbordes (1993)]{d_desb93}
\W\xlink{at198b.txt}{\rawdatadir/at198b.txt}
\W Desbordes (1993) \cite{d_desb93}
ER=1, 81\% Ar

\T at187b - Table~\ref{at187b}~\cite[Desbordes (1988)]{d_desb88}
\W\xlink{at187b.txt}{\rawdatadir/at187b.txt}
\W Desbordes (1988) \cite{d_desb88}
ER=1, 22.2\% Ar

\T at187c - Table~\ref{at187c}~\cite[Desbordes (1988)]{d_desb88}
\W\xlink{at187c.txt}{\rawdatadir/at187c.txt}
\W Desbordes (1988) \cite{d_desb88}
ER=1, 50\% Ar

\T at187d - Table~\ref{at187d}~\cite[Desbordes (1988)]{d_desb88}
\W\xlink{at187d.txt}{\rawdatadir/at187d.txt}
\W Desbordes (1988) \cite{d_desb88}
ER=1, 75\% Ar

\T at69a - Table~\ref{at69a}~\cite[Strehlow (1967)]{r_stre67}
\W\xlink{at69a.txt}{\rawdatadir/at69a.txt}
\W Strehlow (1967) \cite{r_stre67}
ER=1, 85\% Ar

\T at69b - Table~\ref{at69b}~\cite[Strehlow (1967)]{r_stre67}
\W\xlink{at69b.txt}{\rawdatadir/at69b.txt}
\W Strehlow (1967) \cite{r_stre67}
ER=1, 75\% Ar

\T\clearpage

\xname{C2H2-Ox4}
\W\subsubsection{Cell width vs initial pressure; C2H2-O2-Kr}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox4.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox4.gif}}
\T\caption{Cell width vs initial pressure; C2H2-O2-Kr}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox4.ps}

\T at197c - Table~\ref{at197c}~\cite[Desbordes (1993)]{d_desb93}
\W\xlink{at197c.txt}{\rawdatadir/at197c.txt}
\W Desbordes (1993) \cite{d_desb93}
ER=1, 75\% Kr

\T at198c - Table~\ref{at198c}~\cite[Desbordes (1993)]{d_desb93}
\W\xlink{at198c.txt}{\rawdatadir/at198c.txt}
\W Desbordes (1993) \cite{d_desb93}
ER=1, 81\% Kr

\T\clearpage

\xname{C2H2-Ox5}
\W\subsubsection{Cell width vs initial pressure; C2H2-O2-He}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox5.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox5.gif}}
\T\caption{Cell width vs initial pressure; C2H2-O2-He}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox5.ps}

\T at197a - Table~\ref{at197a}~\cite[Desbordes (1993)]{d_desb93}
\W\xlink{at197a.txt}{\rawdatadir/at197a.txt}
\W Desbordes (1993) \cite{d_desb93}
ER=1, 75\% He

\T at198a - Table~\ref{at198a}~\cite[Desbordes (1993)]{d_desb93}
\W\xlink{at198a.txt}{\rawdatadir/at198a.txt}
\W Desbordes (1993) \cite{d_desb93}
ER=1, 81\% He

\T at129f - Table~\ref{at129f}~\cite[Strehlow (1969)]{r_stre69b}
\W\xlink{at129f.txt}{\rawdatadir/at129f.txt}
\W Strehlow (1969) \cite{r_stre69b}
ER=1, 75\% He

\T\clearpage

\xname{C2H2-Ox6}
\W\subsubsection{Cell width vs percent diluent; C2H2-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox6.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox6.gif}}
\T\caption{Cell width vs percent diluent; C2H2-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox6.ps}

\T mk6 - Table~\ref{mk6}~\cite[Murray (1986)]{s_murr86}
\W\xlink{mk6.txt}{\rawdatadir/mk6.txt}
\W Murray (1986) \cite{s_murr86}
P=101.3 kPa, ER=1

\T mk7a - Table~\ref{mk7a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{mk7a.txt}{\rawdatadir/mk7a.txt}
\W Knystautas (1982) \cite{r_knys82}
P=101.3 kPa, ER=1

\T at86a - Table~\ref{at86a}~\cite[Lee (1982)]{j_lee82}
\W\xlink{at86a.txt}{\rawdatadir/at86a.txt}
\W Lee (1982) \cite{j_lee82}
P=101.3 kPa, ER=1

\T\clearpage

\xname{C2H2-Air1}
\W\subsubsection{Cell width vs initial temperature; C2H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Air1.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Air1.gif}}
\T\caption{Cell width vs initial temperature; C2H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Air1.ps}

\T at172a - Table~\ref{at172a}~\cite[Tieszen (1991)]{s_ties91}
\W\xlink{at172a.txt}{\rawdatadir/at172a.txt}
\W Tieszen (1991) \cite{s_ties91}
P=101.325 kPa, ER=1

\T\clearpage

\xname{C2H2-Air2}
\W\subsubsection{Cell width vs equivalence ratio; C2H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Air2.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Air2.gif}}
\T\caption{Cell width vs equivalence ratio; C2H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Air2.ps}

\T mk157n - Table~\ref{mk157n}~\cite[Moen (1984)]{i_moen84}
\W\xlink{mk157n.txt}{\rawdatadir/mk157n.txt}
\W Moen (1984) \cite{i_moen84}
T=293 K, P=92.5 kPa

\T js13c - Table~\ref{js13c}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{js13c.txt}{\rawdatadir/js13c.txt}
\W Knystautas (1984) \cite{r_knys84}
T=293 K, P=101.3 kPa

\T at157l - Table~\ref{at157l}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at157l.txt}{\rawdatadir/at157l.txt}
\W Bull (1982) \cite{d_bull82b}
T=293 K, P=101.3 kPa

\T\clearpage

\xname{C2H2-Air3}
\W\subsubsection{Cell length vs initial pressure; C2H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Air3.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Air3.gif}}
\T\caption{Cell length vs initial pressure; C2H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Air3.ps}

\T at140a - Table~\ref{at140a}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at140a.txt}{\rawdatadir/at140a.txt}
\W Bull (1982) \cite{d_bull82b}
T=293 K, ER=1

\T\clearpage

\xname{C2H4-Air1}
\W\subsubsection{Cell width vs equivalence ratio - part1; C2H4-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Air1.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Air1.gif}}
\T\caption{Cell width vs equivalence ratio - part1; C2H4-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Air1.ps}

\T at203 - Table~\ref{at203}~\cite[Moen (1982)]{i_moen82}
\W\xlink{at203.txt}{\rawdatadir/at203.txt}
\W Moen (1982) \cite{i_moen82}
T=293 K, P=92.5 kPa

\T at157c - Table~\ref{at157c}~\cite[Moen (1984)]{i_moen84}
\W\xlink{at157c.txt}{\rawdatadir/at157c.txt}
\W Moen (1984) \cite{i_moen84}
T=293 K, P=92.5 kPa

\T at13b - Table~\ref{at13b}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{at13b.txt}{\rawdatadir/at13b.txt}
\W Knystautas (1984) \cite{r_knys84}
T=293 K, P=101.3 kPa

\T at157e - Table~\ref{at157e}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at157e.txt}{\rawdatadir/at157e.txt}
\W Bull (1982) \cite{d_bull82b}
T=293 K, P=101.3 kPa

\T\clearpage

\xname{C2H4-Air2}
\W\subsubsection{Cell width vs equivalence ratio - part2; C2H4-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Air2.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Air2.gif}}
\T\caption{Cell width vs equivalence ratio - part2; C2H4-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Air2.ps}

\T at157f - Table~\ref{at157f}~\cite[Moen (1984)]{i_moen84}
\W\xlink{at157f.txt}{\rawdatadir/at157f.txt}
\W Moen (1984) \cite{i_moen84}
T=293 K, P=92.5 kPa

\T mk157g - Table~\ref{mk157g}~\cite[Moen (1984)]{i_moen84}
\W\xlink{mk157g.txt}{\rawdatadir/mk157g.txt}
\W Moen (1984) \cite{i_moen84}
T=293 K, P=92.5 kPa

\T at158 - Table~\ref{at158}~\cite[Murray (1984)]{s_murr84}
\W\xlink{at158.txt}{\rawdatadir/at158.txt}
\W Murray (1984) \cite{s_murr84}
T=293 K, P=101.3 kPa

\T\clearpage

\xname{C2H4-Air3}
\W\subsubsection{Cell length vs initial pressure; C2H4-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Air3.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Air3.gif}}
\T\caption{Cell length vs initial pressure; C2H4-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Air3.ps}

\T at140b - Table~\ref{at140b}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at140b.txt}{\rawdatadir/at140b.txt}
\W Bull (1982) \cite{d_bull82b}


\T mk4a - Table~\ref{mk4a}~\cite[Bauer (1986)]{p_baue86}
\W\xlink{mk4a.txt}{\rawdatadir/mk4a.txt}
\W Bauer (1986) \cite{p_baue86}


\T mk4b - Table~\ref{mk4b}~\cite[Bauer (1986)]{p_baue86}
\W\xlink{mk4b.txt}{\rawdatadir/mk4b.txt}
\W Bauer (1986) \cite{p_baue86}


\T mk4c - Table~\ref{mk4c}~\cite[Bauer (1986)]{p_baue86}
\W\xlink{mk4c.txt}{\rawdatadir/mk4c.txt}
\W Bauer (1986) \cite{p_baue86}


\T mk4d - Table~\ref{mk4d}~\cite[Bauer (1986)]{p_baue86}
\W\xlink{mk4d.txt}{\rawdatadir/mk4d.txt}
\W Bauer (1986) \cite{p_baue86}


\T\clearpage

\xname{C2H4-Ox1}
\W\subsubsection{Cell width vs initial pressure; C2H4-O2-Ar}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Ox1.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Ox1.gif}}
\T\caption{Cell width vs initial pressure; C2H4-O2-Ar}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Ox1.ps}

\T at149a - Table~\ref{at149a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at149a.txt}{\rawdatadir/at149a.txt}
\W Knystautas (1982) \cite{r_knys82}
P=1.3-13.3 kPa

\T at128b - Table~\ref{at128b}~\cite[Strehlow (1969)]{r_stre69b}
\W\xlink{at128b.txt}{\rawdatadir/at128b.txt}
\W Strehlow (1969) \cite{r_stre69b}
P=4.1-15.2 kPa

\T at128c - Table~\ref{at128c}~\cite[Strehlow (1969)]{r_stre69b}
\W\xlink{at128c.txt}{\rawdatadir/at128c.txt}
\W Strehlow (1969) \cite{r_stre69b}
P=4.1-35.5 kPa, 50\% Ar

\T at128d - Table~\ref{at128d}~\cite[Strehlow (1969)]{r_stre69b}
\W\xlink{at128d.txt}{\rawdatadir/at128d.txt}
\W Strehlow (1969) \cite{r_stre69b}
P=12.2-70.9 kPa, 75\% Ar

\T at192b - Table~\ref{at192b}~\cite[Abid (1991)]{s_abid91}
\W\xlink{at192b.txt}{\rawdatadir/at192b.txt}
\W Abid (1991) \cite{s_abid91}
P=1.2-14 kPa

\T\clearpage

\xname{C2H4-Ox2}
\W\subsubsection{Cell width vs initial pressure; C2H4-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Ox2.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Ox2.gif}}
\T\caption{Cell width vs initial pressure; C2H4-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Ox2.ps}

\T ja10a - Table~\ref{ja10a}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja10a.txt}{\rawdatadir/ja10a.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1, 55.6\% N2

\T ja10b - Table~\ref{ja10b}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja10b.txt}{\rawdatadir/ja10b.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1, 42.9\% N2

\T ja10c - Table~\ref{ja10c}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja10c.txt}{\rawdatadir/ja10c.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1, 60\% N2

\T ja10d - Table~\ref{ja10d}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja10d.txt}{\rawdatadir/ja10d.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1, 69.2\% N2

\T ja10e - Table~\ref{ja10e}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja10e.txt}{\rawdatadir/ja10e.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1, 73.8\% N2

\T\clearpage

\xname{C2H4-Ox3}
\W\subsubsection{Cell width vs percent diluent; C2H4-O2-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Ox3.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Ox3.gif}}
\T\caption{Cell width vs percent diluent; C2H4-O2-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Ox3.ps}

\T mk7c - Table~\ref{mk7c}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{mk7c.txt}{\rawdatadir/mk7c.txt}
\W Knystautas (1982) \cite{r_knys82}
P=101.3 kPa, 51 - 74\% N2

\T ja11a - Table~\ref{ja11a}~\cite[Kaneshige (1999)]{m_kane99}
\W\xlink{ja11a.txt}{\rawdatadir/ja11a.txt}
\W Kaneshige (1999) \cite{m_kane99}
P=45-50 kPa, 38.5-63.6\% N2

\T ja11b - Table~\ref{ja11b}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja11b.txt}{\rawdatadir/ja11b.txt}
\W EDL (unpublished) \cite{cit_unpub}
P=50 kPa, 0-73.8\% N2

\T ja12 - Table~\ref{ja12}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja12.txt}{\rawdatadir/ja12.txt}
\W EDL (unpublished) \cite{cit_unpub}
P=50 kPa, 50-88.2\% Ar

\T\clearpage

\xname{C2H6-Air1}
\W\subsubsection{Cell width vs equivalence ratio; C2H6-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H6-Air1.ps,width=6in}}
{\htmlimage{\plotdir/C2H6-Air1.gif}}
\T\caption{Cell width vs equivalence ratio; C2H6-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H6-Air1.ps}

\T at157h - Table~\ref{at157h}~\cite[Moen (1984)]{i_moen84}
\W\xlink{at157h.txt}{\rawdatadir/at157h.txt}
\W Moen (1984) \cite{i_moen84}
T=293 K, P=92.5 kPa, ER=1.0-1.3

\T at13d - Table~\ref{at13d}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{at13d.txt}{\rawdatadir/at13d.txt}
\W Knystautas (1984) \cite{r_knys84}
T=293 K, P=101.3 kPa, ER=0.8 - 1.3

\T at157j - Table~\ref{at157j}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at157j.txt}{\rawdatadir/at157j.txt}
\W Bull (1982) \cite{d_bull82b}
T=293 K, P=101.3 kPa, ER=1

\T\clearpage

\xname{C2H6-Air3}
\W\subsubsection{Cell length vs equivalence ratio; C2H6-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H6-Air3.ps,width=6in}}
{\htmlimage{\plotdir/C2H6-Air3.gif}}
\T\caption{Cell length vs equivalence ratio; C2H6-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H6-Air3.ps}

\T at93a - Table~\ref{at93a}~\cite[Bull (1982)]{d_bull82a}
\W\xlink{at93a.txt}{\rawdatadir/at93a.txt}
\W Bull (1982) \cite{d_bull82a}
T=293 K, P=33.8 kPa, ER=0.5-2, \% N2

\T at93b - Table~\ref{at93b}~\cite[Bull (1982)]{d_bull82a}
\W\xlink{at93b.txt}{\rawdatadir/at93b.txt}
\W Bull (1982) \cite{d_bull82a}
T=293 K, P=67.5 kPa, ER=0.5-2, \% N2

\T\clearpage

\xname{C2H6-Ox1}
\W\subsubsection{Cell length vs initial pressure; C2H6+3.5O2+zN2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H6-Ox1.ps,width=6in}}
{\htmlimage{\plotdir/C2H6-Ox1.gif}}
\T\caption{Cell length vs initial pressure; C2H6+3.5O2+zN2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H6-Ox1.ps}

\T at139a - Table~\ref{at139a}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at139a.txt}{\rawdatadir/at139a.txt}
\W Bull (1982) \cite{d_bull82b}


\T at139b - Table~\ref{at139b}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at139b.txt}{\rawdatadir/at139b.txt}
\W Bull (1982) \cite{d_bull82b}
44\% N2

\T at139c - Table~\ref{at139c}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at139c.txt}{\rawdatadir/at139c.txt}
\W Bull (1982) \cite{d_bull82b}
61\% N2

\T at139d - Table~\ref{at139d}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at139d.txt}{\rawdatadir/at139d.txt}
\W Bull (1982) \cite{d_bull82b}
70\% N2

\T at139e - Table~\ref{at139e}~\cite[Bull (1982)]{d_bull82b}
\W\xlink{at139e.txt}{\rawdatadir/at139e.txt}
\W Bull (1982) \cite{d_bull82b}


\T\clearpage

\xname{C3H8-Ox1}
\W\subsubsection{Cell width vs percent diluent; C3H8-O2-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C3H8-Ox1.ps,width=6in}}
{\htmlimage{\plotdir/C3H8-Ox1.gif}}
\T\caption{Cell width vs percent diluent; C3H8-O2-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C3H8-Ox1.ps}

\T mk7d - Table~\ref{mk7d}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{mk7d.txt}{\rawdatadir/mk7d.txt}
\W Knystautas (1982) \cite{r_knys82}
T=293 K, P=101.3 kPa, ER=1, 44 - 76\% N2

\T ja9 - Table~\ref{ja9}~\cite[Kaneshige (1999)]{m_kane99}
\W\xlink{ja9.txt}{\rawdatadir/ja9.txt}
\W Kaneshige (1999) \cite{m_kane99}
T=293 K, P=50 kPa, ER=1, 40-87\% Ar

\T\clearpage

\xname{C3H8-Ox2}
\W\subsubsection{Cell width vs. initial pressure; C3H8-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C3H8-Ox2.ps,width=6in}}
{\htmlimage{\plotdir/C3H8-Ox2.gif}}
\T\caption{Cell width vs. initial pressure; C3H8-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C3H8-Ox2.ps}

\T ja8a - Table~\ref{ja8a}~\cite[Kaneshige (1999)]{m_kane99}
\W\xlink{ja8a.txt}{\rawdatadir/ja8a.txt}
\W Kaneshige (1999) \cite{m_kane99}
T=293 K, ER=1

\T ja8b - Table~\ref{ja8b}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja8b.txt}{\rawdatadir/ja8b.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1, 45.5\% N2

\T ja8c - Table~\ref{ja8c}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja8c.txt}{\rawdatadir/ja8c.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1, 62.5\% N2

\T ja8d - Table~\ref{ja8d}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja8d.txt}{\rawdatadir/ja8d.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1, 71.4\% N2

\T ja8e - Table~\ref{ja8e}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja8e.txt}{\rawdatadir/ja8e.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1, 75.8\% N2

\T\clearpage

\xname{C3H8-Air}
\W\subsubsection{Cell width vs equivalence ratio; C3H8-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C3H8-Air.ps,width=6in}}
{\htmlimage{\plotdir/C3H8-Air.gif}}
\T\caption{Cell width vs equivalence ratio; C3H8-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C3H8-Air.ps}

\T at157k - Table~\ref{at157k}~\cite[Moen (1984)]{i_moen84}
\W\xlink{at157k.txt}{\rawdatadir/at157k.txt}
\W Moen (1984) \cite{i_moen84}


\T\clearpage

\xname{HC-Ox4}
\W\subsubsection{Cell width vs initial pressure; C2H6+3.5O2, C3H8+5O2, C3H6+4.5O2, C4H10+6.5O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Ox4.ps,width=6in}}
{\htmlimage{\plotdir/HC-Ox4.gif}}
\T\caption{Cell width vs initial pressure; C2H6+3.5O2, C3H8+5O2, C3H6+4.5O2, C4H10+6.5O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Ox4.ps}

\T at152 - Table~\ref{at152}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at152.txt}{\rawdatadir/at152.txt}
\W Knystautas (1982) \cite{r_knys82}
Fuel=C2H6

\T at153 - Table~\ref{at153}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at153.txt}{\rawdatadir/at153.txt}
\W Knystautas (1982) \cite{r_knys82}
Fuel=C3H6

\T at154a - Table~\ref{at154a}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at154a.txt}{\rawdatadir/at154a.txt}
\W Knystautas (1982) \cite{r_knys82}
Fuel=C3H8

\T at154b - Table~\ref{at154b}~\cite[Knystautas (1982)]{r_knys82}
\W\xlink{at154b.txt}{\rawdatadir/at154b.txt}
\W Knystautas (1982) \cite{r_knys82}
Fuel=C4H10

\T\clearpage

\xname{HC-Ox5}
\W\subsubsection{Cell width vs percent diluent; C6H14-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Ox5.ps,width=6in}}
{\htmlimage{\plotdir/HC-Ox5.gif}}
\T\caption{Cell width vs percent diluent; C6H14-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Ox5.ps}

\T ja13 - Table~\ref{ja13}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja13.txt}{\rawdatadir/ja13.txt}
\W EDL (unpublished) \cite{cit_unpub}
P=40 kPa, ER=1

\T\clearpage

\xname{HC-Air1}
\W\subsubsection{Cell width vs equivalence ratio; H2-air, C2H2-air, C2H4-air, C2H6-air,C3H8-air, C4H10-air, CH4-air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Air1.ps,width=6in}}
{\htmlimage{\plotdir/HC-Air1.gif}}
\T\caption{Cell width vs equivalence ratio; H2-air, C2H2-air, C2H4-air, C2H6-air,C3H8-air, C4H10-air, CH4-air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Air1.ps}

\T H2-Air1 - Table~\ref{H2-Air1}~\cite[Guirao (1982)]{c_guir82}
\W\xlink{H2-Air1.txt}{\rawdatadir/H2-Air1.txt}
\W Guirao (1982) \cite{c_guir82}
Fuel=H2

\T js13c - Table~\ref{js13c}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{js13c.txt}{\rawdatadir/js13c.txt}
\W Knystautas (1984) \cite{r_knys84}
Fuel=C2H2

\T at13b - Table~\ref{at13b}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{at13b.txt}{\rawdatadir/at13b.txt}
\W Knystautas (1984) \cite{r_knys84}
Fuel=C2H4

\T at13d - Table~\ref{at13d}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{at13d.txt}{\rawdatadir/at13d.txt}
\W Knystautas (1984) \cite{r_knys84}
Fuel=C2H6

\T at13e - Table~\ref{at13e}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{at13e.txt}{\rawdatadir/at13e.txt}
\W Knystautas (1984) \cite{r_knys84}
Fuel=C3H8

\T at13f - Table~\ref{at13f}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{at13f.txt}{\rawdatadir/at13f.txt}
\W Knystautas (1984) \cite{r_knys84}
Fuel=C4H10

\T at13a - Table~\ref{at13a}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{at13a.txt}{\rawdatadir/at13a.txt}
\W Knystautas (1984) \cite{r_knys84}
Fuel=CH4

\T\clearpage

\xname{HC-Air2}
\W\subsubsection{Cell width vs. percent additive; C6H14-air-H2, C6H14-air-C2H4, C6H14-air-C2H2, C6H14-air-CO}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Air2.ps,width=6in}}
{\htmlimage{\plotdir/HC-Air2.gif}}
\T\caption{Cell width vs. percent additive; C6H14-air-H2, C6H14-air-C2H4, C6H14-air-C2H2, C6H14-air-CO}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Air2.ps}

\T ja14a - Table~\ref{ja14a}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja14a.txt}{\rawdatadir/ja14a.txt}
\W EDL (unpublished) \cite{cit_unpub}
ER=1, 0-26\% H2

\T ja14b - Table~\ref{ja14b}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja14b.txt}{\rawdatadir/ja14b.txt}
\W EDL (unpublished) \cite{cit_unpub}
ER=1, 0-5.6\% C2H4

\T ja14c - Table~\ref{ja14c}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja14c.txt}{\rawdatadir/ja14c.txt}
\W EDL (unpublished) \cite{cit_unpub}
ER=1, 0-6.5\% C2H2

\T ja14d - Table~\ref{ja14d}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja14d.txt}{\rawdatadir/ja14d.txt}
\W EDL (unpublished) \cite{cit_unpub}
ER=1, 0-28.6\% CO 

\T\clearpage

\xname{HC-Air3}
\W\subsubsection{Cell width vs. percent additive; CO-H2-Air, CO-C2H2-Air, CO-C2H4-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Air3.ps,width=6in}}
{\htmlimage{\plotdir/HC-Air3.gif}}
\T\caption{Cell width vs. percent additive; CO-H2-Air, CO-C2H2-Air, CO-C2H4-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Air3.ps}

\T ja15a - Table~\ref{ja15a}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja15a.txt}{\rawdatadir/ja15a.txt}
\W EDL (unpublished) \cite{cit_unpub}
P=100 kPa, ER=1, 0.54-8.7\% H2

\T ja15b - Table~\ref{ja15b}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja15b.txt}{\rawdatadir/ja15b.txt}
\W EDL (unpublished) \cite{cit_unpub}
P=100 kPa, ER=1, 0.29-3.71\% C2H2

\T ja15c - Table~\ref{ja15c}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja15c.txt}{\rawdatadir/ja15c.txt}
\W EDL (unpublished) \cite{cit_unpub}
P=100 kPa, ER=1, 0.28-4.23\% C2H4

\T\clearpage

\xname{HC-Air4}
\W\subsubsection{Cell width vs. percent diluent; NH3-O2-N2, NH3-N2O-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Air4.ps,width=6in}}
{\htmlimage{\plotdir/HC-Air4.gif}}
\T\caption{Cell width vs. percent diluent; NH3-O2-N2, NH3-N2O-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Air4.ps}

\T ja16a - Table~\ref{ja16a}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja16a.txt}{\rawdatadir/ja16a.txt}
\W Akbar (1997) \cite{r_akbar97}
P=66-81 kPa, ER=1, 0-22.2\% N2, Oxidizer=O2

\T ja16b - Table~\ref{ja16b}~\cite[Akbar (1997)]{r_akbar97}
\W\xlink{ja16b.txt}{\rawdatadir/ja16b.txt}
\W Akbar (1997) \cite{r_akbar97}
P=55-75 kPa, ER=1, 0-37.5\% N2, Oxidizer=N2O

\T\clearpage

\xname{HC-Air5}
\W\subsubsection{Cell width vs. percent diluent; C3H6O-O2-Diluent}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Air5.ps,width=6in}}
{\htmlimage{\plotdir/HC-Air5.gif}}
\T\caption{Cell width vs. percent diluent; C3H6O-O2-Diluent}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Air5.ps}

\T ja18a - Table~\ref{ja18a}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja18a.txt}{\rawdatadir/ja18a.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, P=22.5 kPa, ER=1, 0-64.3\% N2

\T ja18b - Table~\ref{ja18b}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja18b.txt}{\rawdatadir/ja18b.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, P=22.5 kPa, ER=1, 0-91.7\% Ar

\T\clearpage

\xname{HC-Air6}
\W\subsubsection{Cell width vs initial pressure; C3H6O-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Air6.ps,width=6in}}
{\htmlimage{\plotdir/HC-Air6.gif}}
\T\caption{Cell width vs initial pressure; C3H6O-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Air6.ps}

\T ja17 - Table~\ref{ja17}~\cite[EDL (unpublished)]{cit_unpub}
\W\xlink{ja17.txt}{\rawdatadir/ja17.txt}
\W EDL (unpublished) \cite{cit_unpub}
T=293 K, ER=1

\T\clearpage

\xname{HC-Air7}
\W\subsubsection{Cell width vs. percent additive; C6H6-H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Air7.ps,width=6in}}
{\htmlimage{\plotdir/HC-Air7.gif}}
\T\caption{Cell width vs. percent additive; C6H6-H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Air7.ps}

\T ja19 - Table~\ref{ja19}~\cite[Knystautas (1998)]{r_knys98}
\W\xlink{ja19.txt}{\rawdatadir/ja19.txt}
\W Knystautas (1998) \cite{r_knys98}
T=293 K, P=101.3 kPa, ER=1

\T\clearpage

